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============================================================================
NOAA OCEAN EXPLORATION AND RESEARCH SITUATION REPORT FOR 10/17/2009
============================================================================

	CRUISE:  EX0909 Leg 3
Mapping Field Trial Hawaiian Islands
	PERSONNEL ON BOARD:

NOAA OER:

1. Mashkoor Malik (ERT Inc.)
2. Kelley Elliott (OER)

3. LTjg Megan Nadeau (OER)

4. Webb Pinner (OER)

Others:

1. Adam Lemire (Intern)
2. Melissa Johnson (Intern)


	DATE/TIME FILED:  10/18/09 0930 HST 
FILED BY: LTjg Megan Nadeau
	

	VESSEL:  NOAA Ship Okeanos Explorer (EX)


	

	GEOGRAPHIC AREA:  Papahanaumokuakea Marine National Monument (PMNM)
	


============================================================================
SUMMARY:  
The ship is working outside the PMNM boundary but inside the boundaries of PMNM reporting area (an area 10 miles wide around PMNM boundary). 
MAPPING:

Transmit board # 16 remains failed. Kongsberg Inc has shipped the replacement board and we should receive, install and test it during the next inport (Oct. 21-Oct.26).
TSG pump was secured due to weather conditions.  The ship was pounding into the waves, which causes the TSG pump to take in air and will eventually burn the motor out. To prevent this from happening, the pump must be secured. Once the seas flatten out, the TSG pump will have to be re-primed by the engineers before it can run properly. This evolution usually takes an hour of the engineers time. We should look into a fix during the next drydock period.
SHIP SENSORS:

SCS issues seem to be solved. System Development Group located in Silver Spring, MD is sending the ship additional hard drives and RAM memory for the database server. They are also contacting Microsoft about the SQL Server problem. – As of now, SCS data seems to be working properly.

TELEPRESENCE:

Pinner continues to document procedures and provide the ship with SOPs for all the work he is accomplishing.  
ENGINEERING:
Stbd small boat took on water while alongside the ship on Wednesday, October 14, 2009 and electrical boxes were affected. Engineers continue to diligently work on this issue, most of the wiring needs to be redone. 
============================================================================
ACTIVITIES:
CTD conducted at 1200. XBT’s conducted at 0000, 0600 and 1800.
COMMENTS:
None 

============================================================================

PLAN OF THE DAY FOR 10/17/09: 
Weather: E wind around 18 kt. Scattered showers, mainly before noon. S swell 4 feet at 16 seconds. Waves around 8 ft. 

EVENTS OF THE DAY:

	0000
	Underway as before 

	0800
	Safety Brief (Bridge)

	1215
	CTD cast (800 m)

	1530
	Ops Brief (forward lounge)


0000, 0600 and 1800 conduct XBT cast
Continue multibeam survey 
On Deck for Monday:
Continue multibeam survey

Continue XBT’s and CTD operations

============================================================================
=========================END OF SITUATION REPORT=========================== 
==========================================================================
